\magnification=\magstep1
\nopagenumbers
\parindent=0pt
{\bf Math 166: Honors Calculus II \hfill Spring 1995}


\centerline{\bf Course Information}

\bigskip
INSTRUCTOR: Dennis M. Snow, 203 Comp/Math Bldg, 631-6227, e-mail: snow.1@nd.edu

\bigskip
OFFICE HOURS: 2--3 MWF, or by appointment

\bigskip
TEXT: T.~Apostol, Calculus, Vol.\ I, 2nd Ed.


\bigskip
EXAMS:  There will be two one-hour exams and a two-hour
final on the following dates:
$$\vbox{\halign{#\ \hfil&#\hfil\cr
Exam I &Thursday, March 2\cr
Exam II &Thursday, April 20\cr
Final &Tuesday,  May 9, 1:45 P.M.\cr
}}$$
Make-up exams will only be given for officially excused absenses.

\bigskip
QUIZZES: There will be a short quiz given on Thursday of every week,
except for exam weeks and the first/Easter week of classes. Each
quiz is worth 10 points.

\bigskip
ASSIGNMENTS: There will be regular assignments due on the class day
after they are assigned, skipping Thursdays: $M\to W$, $W\to F$, $F\to
M$. Points will be deducted for late assignments and may not be
accepted at all.

\bigskip
ATTENDANCE: You are expected to attend every class. Excessive
absenses may result in lowering your grade and even failing the
course.

\bigskip
HONOR CODE: The Honor Code is in effect for all exams, quizzes and
assignments. You are encouraged to work together on the assignments, but
copying in any form or submitting work done by others as your own is a
violation of the Honor Code.

\bigskip
EMAIL: Information about the course and regular grade updates will be
sent to you by e-mail. I encourage you to ask questions through this medium; you will
always receive a written reply.

\bigskip
GRADES:  Grades will be based on a total of 550 points broken down as follows:
$$\vbox{\halign{#\ \hfil&#\ \hfil&#\hfil\cr
Exam I\&II &2 @ 100  &= 200\cr
Final &&= 150\cr
Quizzes &10 @ 10 &= 100\cr
Assignments &(scaled) &= 100\cr
}}$$


\bye
